\begin{table}%t2
 \caption{\label{nearbylistaa2}List of 
nearby SNe used in this work.}
\small%\centerline
{
\begin{tabular}{lccc}
 \hline \hline
  SN name &
  Epoch &
 Bands &
  References \\
 &
  (with respect to $B$ maximum) &
 &
 \\ \hline
1981B   & 0 & {\it UBV} & 1\\
1986G   &  $-$3, $-$1, 0, 1, 2 & {\it BV}  & 2\\
1989B   & $-$5, $-$1, 0, 3, 5 & {\it UBVRI}  & 3, 4\\
1990N   & 2, 7 & {\it UBVRI}  & 5\\
1991M   & 3 & {\it VRI}  & 6\\
1992A   & $-$5, $-$1, 3, 5 & {\it UBVRI} & 7\\
1994D   & $-$5, $-$4, $-$2, 2, 4 & {\it UBVRI}  & 8\\
1996X   & $-$2,  0, 1 & {\it BVRI}  & 9\\
1998aq  &  0, 1, 2, 3, 4, 5 &  {\it UBVRI}  & 10\\
1998bu  & $-$3, $-$2, $-$1 & {\it UBVRI}  & 11\\
1999ee  & $-$4, $-$2, 0, 5 & {\it UBVRI}  & 12\\
2000E   & $-$3, $-$2 & {\it UBVRI}  & 13\\ 
2002bf  &  3 & {\it BVRI}  & 14\\
2002bo  & $-$5, $-$4, $-$3, $-$2, $-$1, 4 & {\it UBVRI}  & 15\\
2002er  & $-$5, $-$4, $-$3, $-$2, $-$1, 0, 2, 4, 5 & {\it UBVRI} & 16, 17 \\
2003cg  & $-$5, $-$1, 1, 4 & {\it UBVRI}  & 18\\
2003du  & $-$4, $-$2, $-$1, 0, 1, 3, 4 & UBVR  & 19, 20\\
2004dt  & $-$5, $-$4, $-$3, $-$2, $-$1, 0, 1, 2, 3, 4 & $\cdots$ & 14, 21\\
2004eo  & $-$3, 2 & {\it UBVRI}  & 22\\
2004S   &  1 & {\it UBVRI} & 23\\
2005am  & $-$1, 2 & {\it UBV}  & 24\\
2005cf  & $-$4, $-$3, $-$1, 0, 4, 5 & {\it UBVRI} & 25, 26\\ 
2005hj  & $-$5, 0, 2, 3, 5 & $\cdots$ & 27\\
2006gz  & $-$4, $-$1 & {\it UBVri}& 28\\
\hline 
\noalign{\smallskip}
\multicolumn{4}{c}{ SNe 91bg-like} \\
\noalign{\smallskip}
\hline
1991bg   & 1, 2 & {\it BVRI}  & 29\\
1999by   & $-$5, $-$4, $-$3, 3, 4, 5 & {\it UBVRI}  & 30\\
\hline 
\noalign{\smallskip}
\multicolumn{4}{c}{ SNe 91T-like} \\
\noalign{\smallskip}
\hline
1991T   & $-$3, 0 & {\it UBVRI}  &  31, 32, 33\\
2000cx  & $-$3, $-$2, 0, 1, 5 & {\it UBVRI}  & 34\\ %
\hline 
\noalign{\smallskip}
\multicolumn{4}{c}{ SNe Ia peculiar} \\
\hline
1997br  & $-$4 & {\it UBVRI}  & 35\\
1997cn  &  3 & {\it UBVRI}  & 36\\ 
1999aa  & $-$3, $-$1 & {\it UBVRI}  & 37\\
1999ac  &  0, 2 & {\it UBVRI}  & 38, 39\\
1999aw  &  3, 5 & {\it BVRI}  & 40\\
2002cx  & $-$4, $-$1 & {\it BVRI}  & 41, 42\\
2005hk  & $-$5, $-$4, 4 & {\it UBVRI}  & 43, 44\\
\hline
\end{tabular}}
\smallskip
  References: (1)~\citet{branch83}; 
 (2)~\citet{phillips87}; (3)~\citet{barbon90}; (4)~\citet{wells94};
(5)~\citet{mazzali93}; (6)~\citet{gomez98}; (7)~\citet{kirshner93};
(8)~\citet{patat96}; (9)~\citet{salvo01}; (10)~\citet{branch03}; 
(11)~\citet{jha99}; (12)~\citet{hamuy02}; (13)~\citet{valentini03}; 
 (14)~\citet{leonard05}; (15)~\citet{benetti04};  
(16)~\citet{pignata04}; (17)~\citet{kotak05};
(18)~\citet{eliasrosa06}; (19)~\citet{anupama05}; 
(20)~\citet{stanishev07}; (21)~\citet{altavilla07}; 
(22)~\citet{pastorello07b}; (23)~\citet{krisciunas07}; 
(24)~\citet{brown05}; 
 (25)~\citet{garavini07b}; (26)~\citet{pastorello07a}; 
(27)~\citet{quimby07}; (28)~\citet{hicken07}; 
(29)~\citet{turatto96}; (30)~\citet{garnavich04}; 
(31)~\citet{jeffery92}; (32)~\citet{phillips92}; 
(33)~\citet{mazzali95}; (34)~\citet{li01}; (35)~\citet{li99}; 
(36)~\citet{turatto98}; (37)~\citet{garavini04}; 
(38)~\citet{garavini05}; (39)~\citet{phillips06}; 
(40)~\citet{strolger02}; (41)~\citet{li03}; (42)~\citet{branch04};
(43)~\citet{phillips07}; (44)~\citet{sahu08}. 
\end{table}